Author's response to reviews

Title: Acute hepatitis and myositis associated with Erythema infectiosum by Parvovirus B19 in an adolescent

Authors:

Maria Koliou (mkoliou@spidernet.com.cy)
Evaggelia Karaoli (drellak@googlemail.com)
Elpidoforos S Soteriades (esoteria@hsph.harvard.ac.edu)
Sylvie Pavlides (yackee@cing.ac.cy)
Stavros Bashiardes (sbash@cing.ac.cy)
Christina G Christodoulou (cchristo@cing.ac.cy)

Version: 2 Date: 22 September 2013

Author's response to reviews: see over
September 23\textsuperscript{rd}, 2013

To: Editor  
BMC Paediatrics

Re: 2291709201017781 Acute hepatitis and myositis associated with Erythema infectiosum by Parvovirus B19 in an adolescent

Dear Editor,

We are pleased to submit our revised manuscript entitled “Acute hepatitis and myositis associated with Erythema infectiosum by Parvovirus B19 in an adolescent” for your consideration.

We would like to thank you for giving us the opportunity to revise and resubmit our manuscript. We have made every effort to accommodate all the editors’ and reviewers’ comments, which indeed were very helpful.

Attached please find a copy of our detailed point-by-point response to the provided comments. In addition, we are pleased on behalf of all co-authors to submit our revised manuscript with yellow highlighted changes.

If you have any further questions or need additional information, please feel free to contact me at +357 99646929 or via e-mail. I would like to thank you very much in advance for your consideration.

Sincerely,

Maria Koliou, MD, PhD  
5 Agiou Symeou St,  
2037 Strovolos  
Nicosia, Cyprus  
Tel: + 357 2240 5000 (w)  
+ 357 2231 5520 (h)  
Fax: + 357 2249 9747  
Email: mkoliou@spidernet.com.cy